Author's response to reviews

Title: Recurrent Clostridium Difficile Infection Treated by Home Fecal Transplantation: A Case Report

Authors:

Pauline S Duke (pduke@mun.ca)
John Fardy (jfardy@mun.ca)

Version: 2 Date: 15 May 2014

Author's response to reviews:

Changes have been made to the manuscript as requested by Carlo Tolentino of the Journal Editorial Office in an email dated May 14, 2014. This was done to conform to the journal style. Thank you for your email and suggested revisions.